Author's response to reviews

Title: Neurosyphilis with dementia and bilateral hippocampal atrophy on brain magnetic resonance imaging

Authors:

Shima Mehrabian (shima_meh@yahoo.com)
Margarita Raycheva (margiraycheva@yahoo.com)
Martina Traykova (martina_t9@hotmail.com)
Tonya Stankova (tonya_stankova@yahoo.com)
Latchezar Penev (penevl@gmail.com)
Olga Grigorova (olgag@abv.bg)
Latchezar Traykov (traykov_l@yahoo.fr)

Version: 4 Date: 2 September 2012

Author's response to reviews:

Dear Prof. Jason R Richardson,

We would like to thank the editorial board of the Journal of BMC Neurology for the interest in our manuscript, and for the constructive comments.

Enclosed Please find the revised version of our manuscript. We sought the assistance of the professional editing service “Edanz” to improve the quality of written English.

We hope these changes adequately address the editorial' requests and thank you again for the opportunity to improve the paper.

Sincerely yours,

Shima Mehrabian